Author's response to reviews

Title: The Nrf2-mediated oxidative stress response pathway is associated with tumor cell resistance to arsenic trioxide across the NCI60 panel

Authors:

- Qian Liu (qliu@email.unc.edu)
- Hao Zhang (hzhang@thehamner.org)
- Lisa Smeester (smesta@unc.edu)
- Fei Zou (fzou@bios.unc.edu)
- Matt Kesic (kesic@email.unc.edu)
- Ilona Jaspers (ilona_jaspers@med.unc.edu)
- Jingbo Pi (JPI@thehamner.org)
- Rebecca C Fry (rfry@unc.edu)

Version: 6 Date: 13 August 2010

Author's response to reviews:

Dear Editors

Thank you for providing the opportunity to correct typos. I found a few in the manuscript and have corrected them. The revised main text is attached.

Best regards,

Rebecca Fry